Author's response to reviews

Title: Assessment of routine surveillance data as a tool to investigate measles outbreaks in Mozambique

Authors:

Jagrati V Jani (j.v.jani@medisin.uio.no)
Ilesh V Jani (ivjani@email.com)
Carolina Araujo (tamigor@tvcabo.co.mz)
Sundeep Sahay (sundeeps@ifi.uio.no)
Jorge Barreto (ivjani@email.com)
Gunnar Bjune (g.a.bjune@samfunnsmed.uio.no)

Version: 5 Date: 30 January 2006

Author's response to reviews: see over
To:
The BioMed Central Editorial Team
http://www.biomedcentral.com/

Dear Sir/Madam,

Thanks you for considering the publication of the manuscript numbered 1365486140635245 entitled “Assessment of routine surveillance data as a tool to investigate measles outbreaks in Mozambique” in your Journal.

Please find below authors’ reply to the editor comments.

We hope that you will find the revisions adequate, and the paper can be taken to a final acceptance and publication in BMC Infectious Disease.

Yours sincerely,

Jagrati V. Jani
**Reply to Editor’s version 4**

Formatting Revisions:

1. Section headings – Removed all the numbering from all headings and sub-headings in the manuscript.

2. All the Figures was cropped, and the figures 2, 3 and 4 the axis was formatted in order to equalise the scale (min 0- max 90).